Author's response to reviews

Title: Variation of cataract surgery costs in four different graded providers of China

Authors:

Jiahua Fang (nmhuman@yahoo.com.cn)
Xinghua Wang (huaxinwang529@yahoo.com.cn)
Zhende Lin (linzhende@21cn.com)
Jihua Yan (infojs@163.com)
Ye Yang (nmfyf@yahoo.com.cn)
Jingbo Li (58375581@qq.com)

Version: 7 Date: 1 September 2010

Author's response to reviews: see over
Dear Sabina:

Thank you very much for your letter dated 31th August 2010. We revised our manuscript according to your suggestions. *In the study, we used the principles outlined in the Helsinki declaration. The study was approved by the ethical committees of Yangtze University, Huazhong University of Science and Technology, Sun Yat-sen University and Jingshan County.*

Thank you for the kind advice.

Best Regards.

Jiahua Fang

E-mail: nmhuman@yahoo.com.cn

All authors: Jiahua Fang, Xinghua Wang, Zhende Lin, Jihua Yan, Ye Yang, Jingbo Li